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Abstract
Whey is a by-product of the cheese industry that, when discharged into the environment, causes the eutrophication of natural water
bodies. Although various alternatives for its utilization have been studied and proposed, a significant portion is still discarded
into the environment. Therefore, the objective of this study was to evaluate the digestion of whey using denitrifying sludge in an
UASB reactor. The UASB reactor was set up and operated for 100 days in two stages. In the first stage (0-50 days), it was fed
with acetate (CH3COO−) until reaching a denitrifying steady state. In the second stage (50-100 days), the reactor was fed with
whey as the carbon source. The reactor was fed with 117 mg NO−3 -N L−1 d−1, maintaining a C/N ratio of 1.8 in both stages. In
the first stage, ECH3COO−C and ENO−3 N of 93.35 ± 4.5% and 97.72 ± 1.24%, respectively, were achieved. In the second stage,
Ewhey-C and EN-NO−3

of 95.69 ± 0.61% and 95.68 ± 1.78%, respectively, were obtained. In both stages, YC-HCO−3
and YN2 were

close to one. No accumulation of secondary metabolites was quantified, indicating that the sludge had the capacity to biodegrade
the organic load of the whey.
Keywords: whey, UASB, dairy industry, digestion, denitrifying sludge.

Resumen
El lactosuero es un subproducto de la industria quesera que, al descargarse al medio ambiente, provoca la eutrofización de cuerpos
de agua natural. A pesar de las diversas alternativas estudiadas para su aprovechamiento, gran parte sigue siendo desechada al
medio ambiente. Por ello el objetivo del presente estudio consistió en evaluar la digestión de lactosuero a través de un lodo
desnitrificante en un reactor UASB. Se operó un reactor UASB por 100 días en dos etapas. En la primera (0-50 días), se alimentó
con CH3COO- hasta alcanzar el estado estacionario desnitrificante. En la segunda etapa (50-100 días), el reactor se alimentó con
lactosuero como fuente de carbono. En ambas etapas, se alimentó con 117 mg NO3

--N L-1 d-1, manteniendo una C/N de 1.8.
En la primera etapa, se obtuvieron EC-CH3COO- y EN-NO3- de 93.35 ± 4.5 y 97.72 ± 1.24 %, respectivamente. En la segunda
etapa se obtuvieron EC-lactosuero y EN-NO3- de 95.69 ± 0.61 y 95.68 ± 1.78 %, respectivamente. En ambas etapas, los YC-HCO3-
y YN2 fueron cercanos a uno. No se cuantificó la acumulación de metabolitos secundarios, por lo que el lodo tuvo la capacidad
de biodegradar la carga orgánica del lactosuero en compuestos inocuos.
Palabras clave: Suero, UASB, industria láctea, digestión, lodo desnitrificante.
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1 Introduction

Whey is a by-product derived from the dairy industry,
characterized by a yellow-greenish color and a sweet-
sour taste. It is composed of proteins, carbohydrates,
vitamins, lipids, enzymes, and minerals (Mahmoudi et
al., 2024; López-Torres, 2024). Annually, up to 190
million tons are produced worldwide, and in most
cases, it is discharged directly into the environment
without any form of treatment (Charalambous et al.,
2023).

The components of whey can vary significantly
according to factors such as the composition of
the milk, the type of cheese, the cheese-making
process, the treatment of the cheese milk, and
the post-fermentation processing of the milk (curd)
(Zhubanova et al., 2023). This by-product contains
milk in amounts ranging from 85% to 95% and
nutrients in about 55% (Mahmoudia et al., 2024).
Additionally, it is characterized by containing (kg
m-3) lactose (0.18–60), fats (0.08–11), proteins (1.4–
100), and minerals (10–189) within pH ranges of
4.6 to 6.4 (Caltzontzin-Rabell et al., 2024; Ramos-
Suárez et al., 2024; Gómez & Sánchez, 2019).
Furthermore, it has been reported to contain B-
group vitamins such as thiamine, pantothenic acid,
riboflavin, pyridoxine, nicotinic acid, cobalamin,
and ascorbic acid concentrations of 2.2 mg mL-1

(Besediuk et al., 2024).
Despite the aforementioned benefits, the presence

of whey in the environment causes severe problems,
making the cheese industry one of the most polluting
sectors within the agri-food industry, as this by-
product still lacks commercial value for its utilization
(Ashton & Syed, 2020; Caballero et al., 2021).
The presence of whey in natural water bodies has
caused various contamination issues, as it increases
the biological oxygen demand (BOD) to between 35
to 60 g L-1, the chemical oxygen demand (COD) to
60 to 70 kg m-3, and the dissolved organic carbon
(DOC) to ranges of 50 to 102 g L-1 (Chourasia et al.,
2022). Additionally, whey has been reported to have
low alkalinity (<0.22 g L-1 CaCO3), high phosphate
content (2.0 to 4.5 g L-1), and inorganic nitrogen (1
g L-1) (Casallas-Ojeda et al., 2024). These factors
contribute to the eutrophication of natural water
bodies, resulting from changes in soil composition
due to alterations in nitrogen and phosphorus cycles
(Casallas-Ojeda et al., 2024). To mitigate these
issues, whey has been used as a dietary supplement
and for ethanol production, among other proposed
alternatives that are still under study. For example,
Trejo et al. (2015) proposed the reuse of whey for
the commercialization of products such as fermented
beverages, ricotta cheese, and petit-suisse-type cheese,

as it contains large amounts of nutrients, proteins,
and solids. Nevertheless, whey discharges have been
estimated at up to 10 billion tons per year, with 50%
of this amount not being repurposed for the production
of new products, resulting in the remainder being
released into the environment (Besediuk et al., 2024;
Cárdenas-Medina et al., 2020; Lech & Holownia et al.,
2015). Consequently, the study of processes capable
of degrading or fully eliminating these contaminants
from water is imperative.

While conventional wastewater treatment
processes have helped mitigate water pollution, they
do not completely eliminate the high organic loads
caused by the presence of whey, resulting in high
levels of COD. However, various technologies for
treating water contaminated with cheese industry
by-products have been studied. Bazrafshan et al.
(2021) investigated the treatment of dairy industry
wastewater in a coagulation system combined with a
membrane bioreactor of metal-organic frameworks.
They reported a removal of 26% and 75% of
the organic matter present in whey wastewater in
the coagulation and activated sludge processes,
respectively, with an initial COD of 10,000 mg L-1.
The total COD reduction in the system was around
98.8%. Additionally, whey was diluted with deionized
water in this system due to the high organic loads
present in the water. Moreover, whey is characterized
by high nitrogen concentrations. Elia et al. (2023)
reported total nitrogen (TN) concentrations ranging
from 340 to 1040 mg L-1, NO−3 levels between 4 to
8 mg L-1, and dissolved organic carbon ranging from
75,300 to 80,300 mg L-1.

In addition to whey pollution, large quantities
of nitrogen are also discarded daily from various
domestic and industrial anthropogenic activities.
Nitrogen is an important component of all living
organisms. However, the presence of species such
as NO−3 and nitrite (NO−2 ) in the environment
can be due to the ecosystem's geology and, in
the worst and most concerning case, anthropogenic
activities. The main sources of NO−3 discharge are
domestic wastewater, agricultural runoff, and the
mineralization of organic nitrogen in soil and water
(Xu et al., 2024). Exposure and ingestion of NO−3
in humans can lead to an increase in methemoglobin
in the blood (methemoglobinemia), reducing oxygen
transport through the blood. According to the World
Health Organization and the European Drinking Water
Directive (98/83/EC, 1998), the threshold value for
NO−3 intake and consumption in drinking water
is 0.8 mM (Margalef-Marti et al., 2019). In the
environment, high nitrogen concentrations increase
the eutrophication of natural water bodies, which
decreases the photosynthetic activity of aquatic plants
and thus reduces the concentration of dissolved
oxygen (Luo et al., 2022).
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Table 1. Composition of the UASB feed sources in the two operational stages.
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2. Could Equation 1, "CH₃COO⁻ + 1.6NO₃⁻ → 2HCO₃⁻ + 0.8N₂ + 0.2H₂O + 0.6OH⁻" be displayed 
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         Table 1. Composition of the UASB feed sources in the two operational stages. 

Composition of the feed 

source 

First 

stage 

(mg L-1) 

Second stage 

 (mg L-1) 

Carbon 

source 

CH3COO--C 1000.00 - 

KH2PO4 1000.00 500.00 

Na2MoO4●2H2O 120.00 60.00 

CaCl2●2H2O 60.00 30.00 

MgSO4●7H2O 400.00 200.00 

Whey-C - 1000 

Nitrogen 

source 

NO3
-N 555.55 510.42 

CuSO4●5H2O 20.00 20.00 

FeCl3●6H2O 400.00 400.00 

5. In Figure 1, center the figure caption and ensure the colors are preserved as specified. They 

should be as follows “Figure 1. Feeding profile of the carbon and nitrogen source to 

the UASB reactor. (●) TOC, (▲) TN, (♦) HCO3
--C” 

6. In Figure 2, center the figure caption and ensure the colors are preserved as specified. They 

should be as follows ”Figure 2. pH evaluation in the UASB reactor fed with acetate and whey. 

(●) pH”  

7. In Figure 3, ensure the colors in the figure caption are preserved as specified. The caption 

should be as follows “Figure 3. Effluent profile of the UASB reactor fed with acetate 

and whey. A) Carbonaceous compounds: (●) TOC, (♦) HCO3
--C, B) Nitrogenous 

compounds: (▲) TN, (■) N2.” 
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and carbon accumulation from secondary metabolites of the denitrification process when 
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There are various chemical, physical, and
biological treatment technologies for the removal of
nitrogen from wastewater, but biological processes are
more commonly used due to their low cost and energy
consumption. These are classified into anaerobic-
anoxic-oxic (AAO) processes, anaerobic/oxic (AO)
processes, which are carried out in sequencing batch
reactors (SBR), moving bed biofilm reactors (MBBR),
upflow anaerobic sludge blanket reactors (UASB),
among others (Fan et al., 2024). Denitrification is a
biological process that consists of the reduction of
nitrate ions (NO−3 ) to gaseous nitrogen (N2), while an
organic source as CH3COO- is oxidized to CO2 and
HCO−3 (Equation 1), depending on working conditions
(Camacho et al., 2023; Martínez et al., 2021).

CH3COO− + 1.6NO−3 → 2HCO−3 + 0.8N2 + 0.2H2O

+0.6OH−

(1)

This process has achieved complete biodegradation
of various recalcitrant compounds from the
petrochemical and pharmaceutical industries in batch
systems to environmentally harmless compounds
(Islas-García et al., 2017). The study of these types
of processes in continuous systems could represent a
sustainable and efficient alternative for the removal
of nitrogen and carbon components present in whey.
Therefore, the objective of this work was to evaluate
the digestion of whey by denitrifying sludge in an
Upflow Anaerobic Sludge Blanket (UASB) reactor.

2 Methodology

2.1 Degradation of Whey in the UASB

A 1.340 L upflow anaerobic sludge blanket (UASB)
reactor was set up and operated, inoculated with 3.58
g L−1 of sludge from the Atotonilco de Tula Water
Treatment Plant, Hidalgo, Mexico. The reactor was
monitored for 100 days in two stages and was fed
with a separate carbon and nitrogen source. In the first

stage (50 days), the reactor was fed with CH3COO−-
C and NO−3 -N at a C/N ratio of 1.8, following the
composition shown in Table 1, until achieving a steady
denitrifying state. In stage 2 (from day 51), the UASB
was fed with whey as the carbon source and NO−3 ,
maintaining the C/N ratio of 1.95 (Table 1). The whey
used to feed the UASB was sourced from an artisanal
cheese industry in Tulancingo de Bravo, Hidalgo.

The UASB operated at a constant temperature of
20°C, with a hydraulic retention time of 2.2 days and a
pH of 9.47 ± 0.25. Liquid and gas samples were taken
at the reactor outlet over the 100-day operation period
for subsequent analysis. Liquid samples at the UASB
outlet were centrifuged for 10 minutes at 5000 rpm,
and the supernatant was filtered through a 0.22 µm
nylon membrane pore size. Biogas production at the
UASB outlet was measured by displacement using an
inverted column fully submerged in a pH 3 saturated
NaCl solution (360 g L-1). Substrate and product
concentrations were evaluated until the UASB reached
steady-state with a coefficient of variation (CV) of less
than 10%.

2.2 Evaluation of UASB

The biodegradation capacity of the sludge in the
UASB reactor was evaluated by calculating substrate
consumption efficiencies (E), as shown in Equation 2.

E =
(
Co −Ci

Co

)
∗ 100 (2)

Where Co is the initial substrate concentration and
Ci is the final substrate concentration. Additionally,
product yields (Y) were calculated using Equation 3.

Y =
CP

Co −Ci
(3)

Where CP is the concentration of the generated
product. Furthermore, a mass balance of carbonaceous
and nitrogenous compounds at the inlet and outlet of
the UASB was conducted to rule out the accumulation
of intermediate compounds and the production of
metabolites from the denitrification process.
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2.3 Analytical methods 155 
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total nitrogen (TN) were evaluated using a Shimadzu total organic carbon analyzer. 157 
Concentrations of N2 and CO2 were evaluated using a Perkin Elmer Autosystem XL gas 158 
chromatograph (GC) equipped with a thermal conductivity detector (TCD) and an HP-159 
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temperatures were set at 60°C, 100°C, and 200°C, respectively. Total suspended solids 163 
(TSS), fixed suspended solids (FSS), and volatile suspended solids (VSS) were evaluated 164 
following the APHA standard (1998). 165 
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The UASB reactor was operated and monitored for 100 days in two stages of 50 days each. 168 
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Figure 1. Feeding profile of the carbon and nitrogen source to the UASB reactor. (•) TOC, (▲) TN, (♦) HCO3
--C.
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Figure 1. Feeding profile of the carbon and nitrogen source to the UASB reactor. (●) TOC, 179 
(▲) TN, (♦) HCO3--C 180 

The UASB reactor was maintained at a temperature of 20 °C during the 100 days of 181 
operation. The pH in the first stage was 9.47 ± 0.25, and when whey was fed, it decreased to 182 
7.46 ± 0.28 as shown in Figure 2. This decrease is attributed to whey being an acidic 183 
byproduct (pH 4.2 – 5.2), causing a two-unit drop in the medium pH (Besediuk et al., 2024). 184 
However, this change did not have negative effects on the denitrifying respiratory process, 185 
as efficient performance has been reported within pH values ranging from 7 to 9 (Camacho 186 
et al., 2023; Islas et al., 2017). Moreover, under these conditions, a predominant production 187 
of HCO₃⁻ is ensured, minimizing CO₂ release (Guerrero-Martínez et al., 2024). 188 
 189 

 190 

Figure 2. pH evaluation in the UASB reactor fed with acetate and whey. (●) pH 191 

3.2 Evaluation of UASB reactor with acetate and whey 192 
The results of the profile of effluents from the UASB reactor fed with acetate and whey are 193 
shown in Figure 3. During the first 50 days of operation, a gradual increase in CH₃COO⁻-C 194 
consumption was observed, reaching an ECH₃COO-C of 93.35 ± 4.5% (Table 2). 195 
Simultaneously, HCO₃⁻-C was generated as the sole product from acetate biodegradation, 196 
achieving a YHCO₃-C of 0.94 ± 0.08. Regarding nitrogenous compounds, a TN removal 197 
efficiency of 97.72 ± 1.24% was achieved, and this was transformed into N₂, resulting in a 198 
YN₂ of 1.30 ± 0.24. A mass balance of carbonaceous and nitrogenous compounds was 199 
conducted, ruling out nitrogen and carbon accumulation from secondary metabolites of the 200 
denitrification process when fed with CH₃COO⁻-, achieving complete denitrification. 201 
Additionally, during the 100 days of UASB reactor operation, the sludge reached and 202 
maintained steady-state denitrification (SSD), as calculations of response variables such as 203 
substrate consumption efficiencies, product generation rates, and variations were less than 204 
10%. 205 
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Figure 2. pH evaluation in the UASB reactor fed with acetate and whey. (•) pH

2.3 Analytical methods

Concentrations of total carbon (TC), total organic
carbon (TOC), inorganic carbon (IC), and total
nitrogen (TN) were evaluated using a Shimadzu total
organic carbon analyzer. Concentrations of N2 and
CO2 were evaluated using a Perkin Elmer Autosystem
XL gas chromatograph (GC) equipped with a thermal
conductivity detector (TCD) and an HP-PLOT/Q
capillary column from Agilent Technologies (USA)
with dimensions of 30 m length, 0.530 mm internal
diameter, and 40 µm particle size. Ultra-high purity
argon was used as the mobile phase at a flow rate
of 10 mL min-1. The oven, injector, and detector
temperatures were set at 60°C, 100°C, and 200°C,
respectively. Total suspended solids (TSS), fixed
suspended solids (FSS), and volatile suspended solids
(VSS) were evaluated following the APHA standard
(1998).

3 Discussion and results

3.1 UASB Reactor Operation

The UASB reactor was operated and monitored for
100 days in two stages of 50 days each. In the first
stage (0-50 days), the reactor was fed with 1000 mg

CH3COO−-C L−1, and in the second stage (51-100
days), whey was added as the sole carbon source, with
the total organic carbon (TOC) concentration adjusted
to 1000 mg L−1. This was because whey contains high
carbon loads, which could modify sludge metabolism,
and the C/N ratio was maintained at 1.8. For the
nitrogen source, the reactor was continuously fed with
117 mg NO−3 -N L−1 d−1, maintaining a C/N ratio
of 1.8 in both stages. The profile of organic loads
from the carbon and nitrogen sources in the UASB is
shown in Figure 1. This figure illustrates that carbon
and nitrogen loads remained constant throughout the
operation.

The UASB reactor was maintained at a
temperature of 20 °C during the 100 days of operation.
The pH in the first stage was 9.47 ± 0.25, and when
whey was fed, it decreased to 7.46 ± 0.28 as shown
in Figure 2. This decrease is attributed to whey
being an acidic byproduct (pH 4.2 – 5.2), causing
a two-unit drop in the medium pH (Besediuk et al.,
2024). However, this change did not have negative
effects on the denitrifying respiratory process, as
efficient performance has been reported within pH
values ranging from 7 to 9 (Camacho et al., 2023;
Islas et al., 2017). Moreover, under these conditions,
a predominant production of HCO−3 is ensured,
minimizing CO2 release (Guerrero-Martínez et al.,
2024).
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Figure 3. Effluent profile of the UASB reactor fed with acetate and whey. A) Carbonaceous compounds: (•) TOC,
(♦) HCO3
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Table 2. Substrate consumption efficiencies and product yields in the UASB reactor fed with acetate and whey.

Stage Efficiencies (%) Yields (Y)

CH3COO−-C Whey-C NO−3 -N HCO−3 -C N2

1 93.35 ± 4.50 - 97.72 ± 1.24 0.98 ± 0.08 1.30 ± 0.24
2 - 95.69 ± 0.61 95.68 ± 1.78 0.70 ± 0.08 1.20 ± 0.08

3.2 Evaluation of UASB reactor with
acetate and whey

The results of the profile of effluents from the UASB
reactor fed with acetate and whey are shown in
Figure 3. During the first 50 days of operation, a
gradual increase in CH3COO−-C consumption was
observed, reaching an ECH3COO-C of 93.35 ± 4.5%
(Table 2). Simultaneously, HCO−3 -C was generated
as the sole product from acetate biodegradation,
achieving a YHCO3-C of 0.94 ± 0.08. Regarding
nitrogenous compounds, a TN removal efficiency
of 97.72 ± 1.24% was achieved, and this was
transformed into N2, resulting in a YN2 of 1.30
± 0.24. A mass balance of carbonaceous and
nitrogenous compounds was conducted, ruling out
nitrogen and carbon accumulation from secondary
metabolites of the denitrification process when fed
with CH3COO−, achieving complete denitrification.
Additionally, during the 100 days of UASB reactor
operation, the sludge reached and maintained steady-
state denitrification (SSD), as calculations of response
variables such as substrate consumption efficiencies,

product generation rates, and variations were less than
10%.

It should be noted that gaseous compounds such
as CH4, H2S, and CO2 were not quantified in
the chromatograms, so the carbon at the inlet was
transformed solely into HCO−3 . This was because the
reactor was operated at C/N ratios and TOC/SO2−

4
ratios of 1.8 and 2.3, respectively. According to Lu
et al. (2015), a TOC/SO2−

4 ratio close to 20 promotes
CH4 production.

In terms of nitrogen compounds, an ETN of 95.68
± 1.78% was achieved, and nitrogen in the influent of
the reactor was converted to N2, resulting in a YN2
of 1.20 ± 0.08. It is important to mention that the
fed whey contained 47.12 mg L-1 of total nitrogen,
which could originate from proteins, amino acids, as
well as ammonia and urea. Tsermoula et al. (2023)
reported concentrations of up to 2576 ± 1133 and 5746
± 371 µM of ammonia and urea in whey, respectively,
as well as non-protein nitrogen compounds such as
creatine, proline, creatinine, among others. Despite
nitrogen in whey not being predominantly NO3

-,
it was converted to N2, suggesting the occurrence
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of anammox processes during reactor operation,
although denitrification was predominant (Cao et
al., 2024). Additionally, nitrogen concentrations
decreased and did not increase, ruling out the
presence of nitrifying and ammonifying bacteria and
indicating a predominance of denitrifying bacteria
(Bucci et al., 2020; Chen et al., 2024). According
to Lech & Trusek (2015), the biodegradation of
organic nitrogen by bacteria is highly valuable during
the biodegradation of whey. This highlights that
the denitrifying respiratory process is a valuable
alternative for whey biodegradation.

Similar results were reported by Xiong et al.
(2024), achieving nitrogen and dissolved organic
carbon (COD) removal efficiencies of 84 ± 3.7% and
81.4 ± 6.1%, respectively, in a partially denitrifying
and anammox reactor. This reactor was fed with
municipal wastewater with nitrogen and carbon loads
of 155.4 ± 12.6 and 186.5 ± 25.9 mg L-1 d-1,
respectively. The study mentions that sludge first
degrades organic nitrogen followed by compounds like
NO3

-. This could be because the energy expenditure
to convert organic nitrogen to N2 is lower than
reducing NO3

- to gaseous nitrogen. Thus, whey was
biodegraded by denitrifying sludge in the UASB
reactor. Moreover, whey has been reported as a
promoter of denitrification (Margalet et al., 2019).

In other studies, such as that by Siqueira et
al. (2022), organic carbon biodegradation efficiencies
ranged from 69% to 81% in a methanogenic UASB
reactor fed with 1834 to 7215.07 mg L-1 of dissolved
organic carbon. These results are lower than those
obtained in the present study, despite Siqueira's
reported hydraulic retention time of 4.4 days, twice
that established in the present study. Additionally,
Siqueira's study evaluated the formation of CH4,
CO2, and H2S, with a daily gas production of 3.7
liters and up to 4.88 L at best, with compositions
of 66.47%, 11.43%, and 53 mg L-1, respectively.
Although methane is a gaseous biofuel, it is considered
a greenhouse gas (GHG), and the presence of H2S
can pose issues for the commercialization or use of
this biogas (Barreto et al., 2024). However, in the
UASB reactor with denitrifying sludge, only N2 and
HCO3

- were produced, considered environmentally
benign, and because no gaseous metabolites like N2O
accumulated, they represent a zero GHG emission
option for wastewater treatment (Conthe et al., 2019).
Mahmoudi et al. (2024) reported removal efficiencies
of 80% for dissolved organic carbon and 90.4% for
total nitrogen in an artificial wetland system using
bulrushes, cattails, papyrus, and finally floating water
lentils (Lemna minor) for whey removal. These results
were lower than those reported in the present study,
despite the use of three plant species in the study
of Mahmoudi. This demonstrates that the denitrifying
process is an efficient alternative for the degradation of

whey-contaminated water, and the products generated
are innocuous to the environment

Conclusions

El denitrifying sludge achieved a degradation
efficiency of 95.69 ± 0.61% and 95.68 ± 1.78% of
the organic carbon and nitrogen load present in whey,
respectively. The reactor products were solely HCO3-
and N2 with yields of 0.70 ± 0.08 and 1.20 ± 0.08,
respectively. Part of the carbon fed into the reactor
was transformed into biomass, yielding 0.20 ± 0.03.
No gaseous secondary metabolites of denitrification
such as NO and N2O, as well as CH4, were generated.
The respiratory denitrifying was not inhibited by
the presence of whey. Complete denitrification was
achieved in the presence of whey in the UASB reactor.
The use of denitrifying sludge could be a promising
future option for treating wastewater from dairy
industries and even municipal wastewater.

Acknowledgments

CONAHCyT project No. 301060 “Desarrollo de un
laboratorio para análisis de agua, suelo, sedimento y
otras muestras ambientales”.

References

APHA. (1998). Standard methods for the
examination of water and wastewater (20th
edn). Washington, DC: American Public Health
Association.

Ashton, K. Y. & Syed, S. H. R. (2020) Functional,
textural, and sensory properties of milk
protein concentrate-based supercritical fluid
extrudates made with acid whey. International
Journal of Food Properties, 23, 708-721. Doi:
https://doi.org/10.1080/10942912.
2020.1753768

Barreto, M. K., Oliveira, C. T. A. & Henry-
Silva, G. G. (2024). Greenhouse gas emission
flux (CO2, CH4, N2O) from marine shrimp
(Litopenaeus vannamei) monoculture cultures
in the Brazilian semi-arid región. Aquaculture,
582, 740536. Doi: https://doi.org/10.
1016/j.aquaculture.2023.740536

Besediuk, V., Yatskov, M., Korchyk, N., Kucherova,
A. & Maletskyi, Z. (2024). Whey - From waste
to a valuable resource. Journal of Agriculture
and Food Research, 18, 101280. Doi: https:
//doi.org/10.1016/j.jafr.2024.101280

6 www.rmiq.org

javascript:void(0)
javascript:void(0)
https://doi.org/10.1080/10942912.2020.1753768
https://doi.org/10.1080/10942912.2020.1753768
https://doi.org/10.1016/j.aquaculture.2023.740536
https://doi.org/10.1016/j.aquaculture.2023.740536
https://doi.org/10.1016/j.jafr.2024.101280
https://doi.org/10.1016/j.jafr.2024.101280


Lanzagorta-Cortes / Revista Mexicana de Ingeniería Química Vol. 24, No. 1(2025) IA24384

Bazrafshan, N., Dadashi-Firouzjaei, M., Elliott,M.
Moradkhani, A. & Rahimpour, A. (2021).
Preparation and modification of low-fouling
ultrafiltration membranas for cheese whey
trarment by membrane bioreactor. Case Studies
in Chemical and Environmental Engineering, 4,
100137. Doi: https://doi.org/10.1016/j.
cscee.2021.100137

Besediuk, V., Yatskov, M., Korchyk, N., Kucherova,
A. & Maletskyi, Z. (2024). Whey - From waste
to a valuable resource. Journal of Agriculture
and Food Research, 18, 101280. Doi: https://
doi.org/10.1016/j.jafr.2024.101280.

Bucci, P., Coppotelli, B., Morelli, I., Zaritzky,
N. & Caravelli, A. (2020). Simultaneous
heterotrophic nitrification and aerobic
denitrification of wastewater in granular reactor:
Microbial composition by next generation
sequencing analysis. Journal of Water Process
Engineering, 36, 101254. Doi: https://doi.
org/10.1016/j.jwpe.2020.101254

Caballero, A., Caballero, P., León, F., Rodríguez-
Morgado, B., Martín, L., Parrado, L., Vaswani,
J. & Ramos-Martín, A. (2021). Conversion
of whey into value-added products through
fermentation and membrane fractionation.
Water (Switzerland), 13, 1623. Doi: https:
//doi.org/10.3390/w13121623

Caltzontzin-Rabell, V., Feregrino-Pérez, A. A. &
Gutiérrez-Antonio, C. (2024). Bio-upcycling
of cheese whey: transforming waste into
raw materials for biofuels and animal feed.
Heliyon, 10, E32700. Doi: https://doi.org/
10.1016/j.heliyon.2024.e32700

Camacho-López, C., Romo-Gómez, C., Otazo-
Sánchez, E., Gómez Aldapa, C. A., García-
Nieto, E. & Juárez-Santacruz, L. (2023). The
pollution effect on the denitrifying capacity
of methanogenic sludges. Water Science &
Technology, 87, 347-353. Doi: https://doi.
org/10.2166/wst.2022.416

Cao, S., Tao, Y., Fang, J., Du, R. & Peng, Y. (2024).
Biological nitrogen removal from nitrate sewage
via novel CANDAN process in continuous-flow
UASB reactor with municipal wastewater as co-
substrate. Chemical Engineering Journal, 488,
150847. Doi: https://doi.org/10.1016/j.
cej.2024.150847

Cárdenas-Medina, K. N., Fajardo-Ortiz, M. C.,
Schettino-Bermúdez, B. S., Meraz-Rodríguez,
M. A. & Castilla-Hernández, P. (2020).
Acidogenesis/methanogenesis from acid cheese
whey in hybrid-UASB reactors. Revista

Mexicana de Ingeniería Química, 19, 463-473.
Doi: https://doi.org/10.24275/rmiq/
IA1420

Casallas-Ojeda, M., Cabeza, I., Sanchez, N.,
Caicedo-Concha, D. M. & Astals, S. (2024).
Cheese whey and dairy manure anaerobic co-
digestion at psychrophilic conditions: Technical
and environmental evaluation. Environmental
Research, 251, 118525. Doi: https://doi.
org/10.1016/j.envres.2024.118525

Charalambous, P., Constantinou, D., Samanides, C.
G. & Vyrides, I. (2023). Enhancing biogas
production from cheese whey using Zero-
Valent Iron: A comparative analysis of batch
and semi-continuous operation modes. Journal
of Environmental Chemical Engineering, 11,
111278. Doi: https://doi.org/10.1016/j.
jece.2023.111278

Chen, H., Yu, M., Hou, L., Zhang, Z. & Yao, J.
(2024). Development of partial denitrification
process in upflow-anaerobic sludge blanket and
effect of electric field on partial denitrification
performance. Bioresource Technology, 394,
130221. Doi: https://doi.org/10.1016/j.
biortech.2023.130221

Chourasia, R., Phukon, L. C., Abedin, M. M.,
Padhi, S., Singh, S. P. & Rai, A. K.
(2022). Whey valorization by microbial and
enzymatic bioprocesses for the production
of nutraceuticals and value-added products.
Bioresource Technology Reports, 19, 101144.
Doi: https://doi.org/10.1016/j.biteb.
2022.101144

Conthe, M., Lycus, P., Arntzen, M. Ø., Ramos
da Silva, A., Frostegård, Å., Bakken, L. R.,
Kleerebezem, R. & van Loosdrecht, M. C.M.
(2019). Denitrification as an N2O sink. Water
Research, 151, 381-387. Doi: https://doi.
org/10.1016/j.watres.2018.11.087

Elia, S., Stylianou, M. & Agapiou, A. (2023).
Combined EC/EO processes for treating goat
cheese whey wastewater. Sustainable Chemistry
and Pharmacy, 32, 100963. Doi: https://
doi.org/10.1016/j.scp.2022.100963

Gómez-Soto, J. A. & Sánchez-Toro, O. J.
(2019) Producción de galactooligosacáridos:
alternativa para el aprovechamiento del
lactosuero: una revisión. Ingeniería y
Desarrollo, 37, 130-158. Doi: http://dx.
doi.org/10.14482/inde.37.1.637

Guerrero-Martínez, Y. A., Romo-Gómez, C.,
Camacho-López, C., Acevedo-Sandoval, O.A.,

www.rmiq.org 7

https://doi.org/10.1016/j.cscee.2021.100137
https://doi.org/10.1016/j.cscee.2021.100137
https://doi.org/10.1016/j.jafr.2024.101280
https://doi.org/10.1016/j.jafr.2024.101280
https://doi.org/10.1016/j.jwpe.2020.101254
https://doi.org/10.1016/j.jwpe.2020.101254
https://doi.org/10.3390/w13121623
https://doi.org/10.3390/w13121623
https://doi.org/10.1016/j.heliyon.2024.e32700
https://doi.org/10.1016/j.heliyon.2024.e32700
https://doi.org/10.2166/wst.2022.416
https://doi.org/10.2166/wst.2022.416
https://doi.org/10.1016/j.cej.2024.150847
https://doi.org/10.1016/j.cej.2024.150847
https://doi.org/10.24275/rmiq/IA1420
https://doi.org/10.24275/rmiq/IA1420
https://doi.org/10.1016/j.envres.2024.118525
https://doi.org/10.1016/j.envres.2024.118525
https://doi.org/10.1016/j.jece.2023.111278
https://doi.org/10.1016/j.jece.2023.111278
https://doi.org/10.1016/j.biortech.2023.130221
https://doi.org/10.1016/j.biortech.2023.130221
https://doi.org/10.1016/j.biteb.2022.101144
https://doi.org/10.1016/j.biteb.2022.101144
https://doi.org/10.1016/j.watres.2018.11.087
https://doi.org/10.1016/j.watres.2018.11.087
https://doi.org/10.1016/j.scp.2022.100963
https://doi.org/10.1016/j.scp.2022.100963
http://dx.doi.org/10.14482/inde.37.1.637
http://dx.doi.org/10.14482/inde.37.1.637


Lanzagorta-Cortes / Revista Mexicana de Ingeniería Química Vol. 24, No. 1(2025) IA24384

González-Ramírez, C. A. & Montiel-Palma, S.
(2024). HCO3

- production from 17β-estradiol
oxidation by photo-Fenton as a strategy to avoid
the generation of greenhouse gases. Revista
Mexicana de Ingeniería Química, 23, IA24118.
Doi: https://doi.org/10.24275/rmiq/
IA24118

Islas-García, I., Romo-Gómez, C. & Cuervo-López,
F. de M. (2017). Ampicillin Mineralization
by Denitrifying Process: Kinetic and
Metabolic Effects. Applied Biochemistry and
Biotechnology, 183, 1049–1061. Doi: https:
//doi.org/10.1007/s12010-017-2483-7

Lech, M. & Trusek-Holownia, A. (2015).
Biodegradation of whey waste in a continuous
stirred-tank bioreactor. Environment Protection
Engineering, 41, 97-107. Doi: https://doi.
org/10.5277/epe150408

Lopez-Torres, J. (2024) Use of whey in dulce de
leche manufacturing. Sweetened Concentrated
Milk Products: Science, Technology an
Engineering, 11, 181-190. Doi: https://
doi.org/10.1016/B978-0-12-823373-3.
00008-5

Lu, X., Zhen, G., Estrada, A.L., Chen, M.,
Nic, J., Hojo, T., Kubota, K. & Li, Y-
Y. (2015). Operation performance and
granule characterization of upflow anaerobic
sludge blanket (UASB) reactor treating
wastewater with starch as the sole carbon
source. Bioresource Technology, 180, 264-
273. Doi: http://dx.doi.org/10.1016/j.
biortech.2015.01.010

Luo, J., Miao, S., Koju, R., Joshi, T. P.,
Liu, R., Liu, H. & Qu, J. (2022).
Simultaneous removal of aromatic pollutants
and nitrate at high concentrations
by hypersaline denitrification:Long-term
continuous experiments investigation. Water
Research, 216, 118292. Doi: https://doi.
org/10.1016/j.watres.2022.118292

Mahmoudia, A., Hannachia, C., Mhirib, F. &
Hamrouni, B. (2024). Performances of
constructed wetland system to treat whey and
dairy wastewater during a macrophytes life
cycle. Desalination and Water Treatment, 318,
100364. Doi: https://doi.org/10.1016/j.
dwt.2024.100364

Mahmoudi, A., Hannachi, C., Mhiri, F. & Hamrouni,
B. (2024). Performances of constructed wetland
system to treat whey and dairy wastewater
during a macrophytes life cycle. Desalination
and Water Treatment, 318, 100364. Doi:

https://doi.org/10.1016/j.dwt.2024.
100364

Margalef-Marti, R., Carrey, R., Soler, A. &
Otero, N. (2019). Evaluating the potential
use of a dairy industry residue to induce
denitrification in polluted water bodies: A flow-
through experiment. Journal of Environmental
Management, 245, 86-94. Doi: https://doi.
org/10.1016/j.jenvman.2019.03.086

Martínez, C., Sauvage, S., Al, A., Verde, P.A.,
Vörösmarty, C.J. & Sánchez, J.M. (2021).
Desnitrification in wetlands: A review towards
a quiantification at global scale. Science of
The Total Environment, 754, 142398. Doi:
https://doi.org/10.1016/j.scitotenv.
2020.142398

Ramos-Suarez, J. L., Alvarez-Méndez, S. J., Ritter,
A., Mata-González, J. & Camacho Pérez, A.
(2024). A comprehensive evaluation of cheese
whey to produce biogas in the Canary Islands.
Biomass and Bioenergy, 180, 107008. Doi:
https://doi.org/10.1016/j.biombioe.
2023.107008

Siqueira, J. C., Assemany, P. & Siniscalchi, L. A. B.
(2022). Microbial dynamics and methanogenic
potential of co-digestion of sugarcane vinasse
and dairy secondary effluent in an upflow
anaerobic sludge blanket reactor. Bioresource
Technology, 361, 127654. Doi: https://doi.
org/10.1016/j.biortech.2022.127654

Trejo-Trejo, E., Trejo-Trejo, N. & Zuñiga-Morales,
J. (2015). Propuesta para el aprovechamiento de
lactosuero en el Valle del Mezquital. Revista de
Tecnología e Innovación, 2, 581-594.

Tsermoula, P., Bechshøft, R. M., Friis, C., Engelsen,
S. B. & Khakimov, B. (2023). Screening of
non-protein nitrogen compounds in lactose
refining streams from industrial whey permeate
processing. Food Chemistry, 405, 134716. Doi:
https://doi.org/10.1016/j.foodchem.
2022.134716

Xiong, L., Li, X., Li, J., Zhang, Q., Zhang, L.,
Wu, Y. & Peng, Y. (2024). Efficient nitrogen
removal from real municipal wastewater
and mature landfill leachate using partial
nitrification-simultaneous anammox and partial
denitrification process. Water Research, 251,
121088. Doi. https://doi.org/10.1016/j.
watres.2023.121088

Xu, C., Shi, B., Jia, Z., Liu, D., Hu, W., Feng,
C. & Li, R. (2024). Tracing the impacts of
ecological water replenishment on the sources

8 www.rmiq.org

https://doi.org/10.24275/rmiq/IA24118
https://doi.org/10.24275/rmiq/IA24118
https://doi.org/10.1007/s12010-017-2483-7
https://doi.org/10.1007/s12010-017-2483-7
https://doi.org/10.5277/epe150408
https://doi.org/10.5277/epe150408
https://doi.org/10.1016/B978-0-12-823373-3.00008-5
https://doi.org/10.1016/B978-0-12-823373-3.00008-5
https://doi.org/10.1016/B978-0-12-823373-3.00008-5
http://dx.doi.org/10.1016/j.biortech.2015.01.010
http://dx.doi.org/10.1016/j.biortech.2015.01.010
https://doi.org/10.1016/j.watres.2022.118292
https://doi.org/10.1016/j.watres.2022.118292
https://doi.org/10.1016/j.dwt.2024.100364
https://doi.org/10.1016/j.dwt.2024.100364
https://doi.org/10.1016/j.dwt.2024.100364
https://doi.org/10.1016/j.dwt.2024.100364
https://doi.org/10.1016/j.jenvman.2019.03.086
https://doi.org/10.1016/j.jenvman.2019.03.086
https://doi.org/10.1016/j.scitotenv.2020.142398
https://doi.org/10.1016/j.scitotenv.2020.142398
https://doi.org/10.1016/j.biombioe.2023.107008
https://doi.org/10.1016/j.biombioe.2023.107008
https://doi.org/10.1016/j.biortech.2022.127654
https://doi.org/10.1016/j.biortech.2022.127654
https://doi.org/10.1016/j.foodchem.2022.134716
https://doi.org/10.1016/j.foodchem.2022.134716
https://doi.org/10.1016/j.watres.2023.121088
https://doi.org/10.1016/j.watres.2023.121088


Lanzagorta-Cortes / Revista Mexicana de Ingeniería Química Vol. 24, No. 1(2025) IA24384

and transformation of groundwater nitrate
through isotope and microbial análisis. Science
of the Total Environment, 928, 172248. Doi:
https://doi.org/10.1016/j.scitotenv.
2024.172248

Zhubanova, A., Akimbekov, N.,Tastambek, K.,

Abdieva, G., Ualieva, P., Malik, A., Altynbay,
N. & Xiaohui, Q. (2024). Whey to Ethanol:
Unlocking the Potential of Cheese Whey for
Sustainable Bioethanol Production. Engineered
Science, 27, 1-19. Doi: https://dx.doi.
org/10.30919/es998

www.rmiq.org 9

https://doi.org/10.1016/j.scitotenv.2024.172248
https://doi.org/10.1016/j.scitotenv.2024.172248
https://dx.doi.org/10.30919/es998
https://dx.doi.org/10.30919/es998

	Introduction
	Methodology
	Degradation of Whey in the UASB
	Evaluation of UASB
	Analytical methods

	Discussion and results
	UASB Reactor Operation
	Evaluation of UASB reactor with acetate and whey


